University of Massachusetts Medical School eScholarship@UMMS

Open Access Articles

Open Access Publications by UMMS Authors

2016-02-04

Incidence of Dengue Virus Infection in Adults and Children in a Prospective Longitudinal Cohort in the Philippines

Maria Theresa Alera Philippines-AFRIMS Virology Research Unit

Et al.

Let us know how access to this document benefits you.

Follow this and additional works at: https://escholarship.umassmed.edu/oapubs

Part of the Clinical Epidemiology Commons, Epidemiology Commons, Immunology and Infectious Disease Commons, International Public Health Commons, and the Virus Diseases Commons

Repository Citation

Alera MT, Srikiatkhachorn A, Velasco JM, Tac-An IA, Lago CB, Clapham H, Fernandez S, Levy JW, Thaisomboonsuk B, Klungthong C, Macareo LR, Nisalak A, Hermann LL, Villa D, Yoon I. (2016). Incidence of Dengue Virus Infection in Adults and Children in a Prospective Longitudinal Cohort in the Philippines. Open Access Articles. https://doi.org/10.1371/journal.pntd.0004337. Retrieved from https://escholarship.umassmed.edu/oapubs/2759

Creative Commons License

This work is licensed under a Creative Commons 1.0 Public Domain Dedication. This material is brought to you by eScholarship@UMMS. It has been accepted for inclusion in Open Access Articles by an authorized administrator of eScholarship@UMMS. For more information, please contact Lisa.Palmer@umassmed.edu.



Citation: Alera MT, Srikiatkhachorn A, Velasco JM, Tac-An IA, Lago CB, Clapham HE, et al. (2016) Incidence of Dengue Virus Infection in Adults and Children in a Prospective Longitudinal Cohort in the Philippines. PLoS Negl Trop Dis 10(2): e0004337. doi:10.1371/journal.pntd.0004337

Editor: Marilia Sá Carvalho, Oswaldo Cruz Foundation, BRAZIL

Received: July 7, 2015

Accepted: December 7, 2015

Published: February 4, 2016

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This research was funded by a grant from the Armed Forces Health Surveilance Center-Global Emerging Infections Surveillance and Response System (Grant P0140_14_AF) and a Canadian Institutes of Health Research Fellowship for LH. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE**

Incidence of Dengue Virus Infection in Adults and Children in a Prospective Longitudinal Cohort in the Philippines

Maria Theresa Alera¹*, Anon Srikiatkhachorn², John Mark Velasco¹, Ilya A. Tac-An³, Catherine B. Lago¹, Hannah E. Clapham⁴, Stefan Fernandez⁵, Jens W. Levy⁵, Butsaya Thaisomboonsuk⁵, Chonticha Klungthong⁵, Louis R. Macareo⁵, Ananda Nisalak⁵, Laura Hermann^{5,6}, Daisy Villa³, In-Kyu Yoon^{5,7}

 Philippines-AFRIMS Virology Research Unit, CAP Building, Cebu City, Philippines, 2 Division of Infectious Diseases and Immunology, Department of Medicine, University of Massachusetts Medical School, Worcester, Massachusetts, United States of America, 3 Cebu City Health Department, Cebu City, Philippines, 4 Department of Epidemiology, Johns Hopkins School of Public Health, Baltimore, Maryland, United States of America, 5 Department of Virology, Armed Forces Research Institute of Medical Sciences, Bangkok, Thailand, 6 Department of Medicine, University of Toronto, Toronto, Ontario, Canada, 7 Dengue Vaccine Initiative, International Vaccine Institute, Seoul, Korea

* MariaTheresa.Alera.ca@afrims.org

Abstract

Background

The mean age of dengue has been increasing in some but not all countries. We sought to determine the incidence of dengue virus (DENV) infection in adults and children in a prospective cohort study in the Philippines where dengue is hyperendemic.

Methodology/Principal Findings

A prospective cohort of subjects \geq 6 months old in Cebu City, Philippines, underwent active community-based surveillance for acute febrile illnesses by weekly contact. Fever history within the prior seven days was evaluated with an acute illness visit followed by 2, 5, and 8-day, and 3-week convalescent visits. Blood was collected at the acute and 3-week visits. Scheduled visits took place at enrolment and 12 months that included blood collections. Acute samples were tested by DENV PCR and acute/convalescent samples by DENV IgM/ IgG ELISA to identify symptomatic infections. Enrolment and 12-month samples were tested by DENV hemagglutination inhibition (HAI) assay to identify subclinical infections. Of 1,008 enrolled subjects, 854 completed all study activities at 12 months per-protocol undergoing 868 person-years of surveillance. The incidence of symptomatic and subclinical infections was 1.62 and 7.03 per 100 person-years, respectively. However, in subjects >15 years old, only one symptomatic infection occurred whereas 27 subclinical infections were identified. DENV HAI seroprevalence increased sharply with age with baseline multitypic HAIs associated with fewer symptomatic infections. Using a catalytic model, the historical infection rate among dengue naïve individuals was estimated to be high at 11–22%/year.

Competing Interests: The authors have declared that no competing interests exist.

Conclusions/Significance

In this hyperendemic area with high seroprevalence of multitypic DENV HAIs in adults, symptomatic dengue rarely occurred in individuals older than 15 years. Our findings demonstrate that dengue is primarily a pediatric disease in areas with high force of infection. However, the average age of dengue could increase if force of infection decreases over time, as is occurring in some hyperendemic countries such as Thailand.

Author Summary

The average age of dengue has been increasing in some but not all dengue endemic countries. To investigate the age pattern of dengue in people of all ages ≥ 6 months old, a prospective community-based cohort study was undertaken in Cebu City, Philippines where dengue virus has been circulating for many decades. Active surveillance for acute fevers was performed, and acute/convalescent blood samples were tested for evidence of symptomatic dengue. Blood was also collected at enrolment and one year later, and tested serologically to identify subclinical infections. Overall, 1.62 symptomatic and 7.03 subclinical infections per 100 person-years of surveillance were detected. Among people older than 15 years, only one symptomatic dengue case occurred while 27 subclinical infections were identified. By analyzing age-specific dengue serology data, the historical infection rate among people with no prior dengue virus infection was found to be high at around 11–22% per year. Our results show that dengue is primarily a childhood disease in endemic settings where the historical infection rate has been high. However, the average age of dengue could increase if the infection rate decreases over time as is happening in some endemic countries like Thailand.

Introduction

Dengue virus (DENV) is the leading cause of vector-borne viral disease globally with an estimated 390 million infections and 96 million symptomatic cases occurring annually [1]. Ecologic and demographic changes are thought to be major contributing factors to the emergence of dengue over the past few decades [2]. DENV infection can present as asymptomatic infection, subclinical infection, undifferentiated fever, dengue fever, dengue hemorrhagic fever (DHF) with or without dengue shock syndrome (DSS), and other severe forms of dengue [3]. Estimates of the proportion of subclinical infections among all DENV infections have ranged widely depending on the year, location, population, and surveillance method [4]. Prospective longitudinal cohort studies can characterize the burden and clinical spectrum of DENV infections including subclinical infections [5].

Multiple exposures to DENV are generally presumed to result in protective immunity leading to lower incidence of clinically overt disease in adults in areas with hyperendemic transmission [6,7]. However, when dengue does occur in adults, the clinical manifestations may be more apparent than in children, perhaps due to differing physiologies and more frequent comorbid conditions in adults [8–11]. At the same time, some dengue hyperendemic countries have reported an increase in the mean age of dengue [12–15]. An important contributing factor to this age increase may be demographic transition in some countries where decreasing birth and death rates may lead to decreasing force of infection (FOI) due to fewer susceptible individuals entering the population [16,17]. Yet, very few prospective longitudinal cohort studies undergoing active surveillance have been conducted in adults to assess overall incidence and disease burden [18,19]; and even fewer have evaluated dengue incidence and relative proportion of subclinical infections in adults and children within the same cohort.

In the Philippines, a dengue outbreak was first reported as early as 1906 [20], and the first epidemic of severe dengue was documented in Manila in 1953 [21]. Since then, dengue has been hyperendemic in most areas of the country with a general increase in the number of dengue cases over time [22]. We conducted a prospective longitudinal cohort study in Cebu City, Philippines, among subjects of all ages ≥ 6 months, with the objective of determining the age-stratified incidence of DENV infection and the ratio of symptomatic to subclinical infection in adults and children. We found that in a setting with high force of infection with consequently high DENV seroprevalence in adults, symptomatic infections were rare in individuals older than 15 years with almost all infections in this age group being subclinical.

Methods

Ethics Statement

The study was approved by the Institutional Review Boards of Vicente Sotto Memorial Medical Center (VSMMC) in Cebu City, Philippines, and the Walter Reed Army Institute of Research (WRAIR). Written informed consent was obtained from subjects ≥ 18 years old and from parents of subjects < 18 years old. Written assent was obtained from children ≥ 12 years old.

Study Location

The study was conducted in the barangay (i.e., village equivalent) of Punta Princesa, an urban community in Cebu City, capital of Cebu province located within the Visayas region of the Philippines. Punta Princesa encompasses 0.96 sq km with a total population of about 27,000 people. Public outpatient care is available to local residents at the Punta Princesa Health Center with patients who require higher level medical care or hospitalization referred to Cebu City Medical Center, the tertiary public hospital of Cebu City, or to VSMMC, the tertiary public hospital of Cebu province.

Prospective Cohort

Study enrolment and surveillance procedures have been previously described [23]. Subjects were enrolled from March to May 2012 in a prospective longitudinal seroepidemiological cohort study using convenience sampling within the community. Inclusion criteria included age ≥ 6 months and residence within Punta Princesa. Exclusion criteria included known active pulmonary tuberculosis, in order to decrease risk to study staff. Enrolment was limited to one subject per household. Approximately 1,000 subjects were targeted for enrolment with roughly equal distribution among five different age groups: 6 months-5 years, 6–15 years, 16–30 years, 31–50 years, and >50 years old. At enrolment and at 12 months, subjects were administered demographic and health questionnaires, and underwent blood collections. Blood was processed into serum aliquots within 24 hours of phlebotomy, frozen at approximately -70°C, and eventually shipped on dry ice to the Armed Forces Research Institute of Medical Sciences (AFRIMS) in Bangkok, Thailand for testing. The serum was tested by hemagglutination inhibition (HAI) assay for all four DENV serotypes and Japanese encephalitis virus (JEV).

Active Surveillance

Enrolled subjects underwent active community-based surveillance to detect acute febrile illness which was defined as reported fever or measured temperature >38.0°C. Subjects were

instructed to report any fevers, and were contacted by study staff once a week through telephone, short message service, and/or home visits. A history of fever within the prior seven days triggered an acute fever investigation consisting of an acute illness visit followed by subsequent visits at 2, 5 and 8 days, and a convalescent visit at 3 weeks. Clinical assessments were performed at all visits, and blood was collected at the acute and 3-week convalescent visits. Blood was processed into serum aliquots within 24 hours of phlebotomy, frozen at approximately -70°C and shipped on dry ice to AFRIMS in Bangkok, Thailand. Acute serum was tested by hemi-nested reverse transcriptase polymerase chain reaction (PCR) to detect DENV RNA. Paired acute/convalescent sera were tested by an in-house DENV IgM/IgG capture enzymelinked immunosorbent assay (ELISA) and by DENV/JEV HAI.

Laboratory Assays

Qualitative DENV RT-PCR/nested PCR. Viral RNA was extracted from 140 μ l of serum using viral RNA extraction kit (Qiagen, Valencia, CA, USA). A two-step nested PCR was performed as previously described [24]. The first RT-PCR step used consensus sense and antisense primers covering DENV serotypes 1–4. The RT-PCR product was then re-amplified in a second step containing five primers: the original sense primer and four internal, serotype-specific, anti-sense primers.

DENV/JEV IgM/IgG capture ELISA. An in-house anti-DENV/JEV IgM/IgG capture ELISA was performed as previously described [25]. Sucrose-acetone extracted viral antigen passaged in suckling mouse brains was utilized. IgM \geq 40 units indicated an acute DENV infection. IgM-to-IgG ratio \geq 1.8 was considered to be acute primary infection while a ratio <1.8 was considered as acute secondary infection. For paired samples, a two-fold increase in IgG with \geq 100 units was considered to be an acute secondary infection even in the absence of IgM \geq 40 units.

DENV/JEV hemagglutination inhibition assay. HAI serology was performed on acetone-extracted serum using the procedure of Clarke and Casals modified for 96-well V-bottom plates as previously described [26]. Suckling mouse brain passaged, sucrose-extracted viral antigen was added to 25 ul of two-fold serial serum dilutions (1:10 to 1:1280). A \geq four-fold rise in HAI titer for any DENV serotype was considered as a positive seroconversion.

Study Definitions

Acute symptomatic DENV infection was defined as an acute febrile illness with positive DENV PCR in the acute sample, and/or positive DENV IgM/IgG ELISA in the paired acute/convalescent samples. Subclinical DENV infection was defined as a \geq four-fold rise in DENV HAI titer for any serotype in paired enrolment/12-month samples with no acute symptomatic DENV infection detected during the intervening surveillance period.

Subjects with an HAI titer <10 to all four serotypes were considered to be dengue naive. Those with an HAI titer \geq 10 to only one DENV serotype were considered to have monotypic dengue immune status. Those with HAI titers \geq 10 to two or more serotypes were considered to have multitypic immune status.

Statistical Analysis

Descriptive statistics for infection rates, symptoms and other characteristics were performed. To inform the underlying infection dynamics in this population, we employed the chi-square or Fisher's exact test as appropriate to evaluate the association between DENV infection and both age >15 years and the nominal categorical baseline immune status. To estimate FOI, age-stratified DENV HAI seroprevalence data at enrolment was analyzed such that HAI titer <10

for all four DENV serotypes indicated a dengue naive subject while HAI titer ≥ 10 for at least one DENV serotype indicated a non-naive subject. A catalytic model [16,27] was used to estimate FOI (λ) for each DENV HAI serotype assuming independence between serotypes (details available in supplementary information). In this catalytic model, (1—exp [- λ]) was the proportion of the dengue naive population infected each year by each serotype. FOI was estimated in a Markov Chain Monte Carlo (MCMC) framework. Means and 95% credible intervals were reported. When using seroprevalence at a single time point, age and time could have been confounded so that changes in FOI during the last five years, for example, could have been due either to differential exposure in individuals <5 years old or to different FOI in the last five years. Two models with different assumptions were assessed; one in which FOI was assumed to be constant over all years (model 1), and one in which FOI could be different during the previous five years (or equivalently, among individuals <5 years old) from the years before then (model 2). For each model, an average FOI was estimated across all serotypes. Ages were rounded down so those <1 year old were shown as 0. All analyses were performed using R version 3.0.2 (R Foundation for Statistical Computing, Vienna, Austria).

Results

A total of 1,008 subjects were enrolled from March to May 2012. A description of the enrolled cohort is presented in <u>Table 1</u>. During 985 person-years of active surveillance from enrolment to 12 months, 274 acute febrile illnesses were detected with 268 acute and 261 convalescent blood samples collected (Fig 1). Of 50,125 weekly contacts for the 1008 enrolled subjects, only 1.2%

Characteristic	Enrolled, n (%)	Per-protocol, n (%)*
Total subjects	1008 (100)	854 (100)
Age group		
6 mos—5 yrs	203 (20.1)	148 (17.3)
6–15 yrs	201 (19.9)	184 (21.5)
16–30 yrs	200 (19.8)	168 (19.7)
31–50 yrs	204 (20.2)	172 (20.1)
>50 yrs	200(19.8)	182 (21.3)
Male gender	500 (49.6)	415 (48.6)
Number in household		
1	16 (1.6)	15 (1.8)
2–3	207 (20.5)	174 (20.4)
4–6	526 (52.2)	439 (51.4)
7–10	237 (23.5)	205 (24.0)
>10	22 (2.2)	21 (2.5)
Number of children in household		
0	199 (19.7)	172 (20.1)
1	231 (22.9)	190 (22.2)
2	229 (22.7)	188 (22.0)
3	180 (17.9)	156 (18.3)
>3	169 (16.8)	148 (17.3)

Table 1. D	escription	of cohort	subjects.
------------	------------	-----------	-----------

*Per-protocol subjects completed all study activities at 12 months including enrollment and 12-month blood collections.

n = number.

doi:10.1371/journal.pntd.0004337.t001



Fig 1. Study flow chart.

doi:10.1371/journal.pntd.0004337.g001

were unsuccessful due to the subject being "unavailable." These weekly contacts were completed by household visit (22.4%), SMS (37.6%) or telephone call (38.8%). Sixteen acute symptomatic DENV infections were identified during this surveillance period. Thirteen of the 16 symptomatic infections were PCR-positive with serotype data available: 10 DENV-1, two DENV-2 and one DENV-3. A total of 854 subjects completed all study activities by the 12-month visit constituting the "per-protocol" subjects. Of the 154 subjects who did not complete all study activities, 66 relocated out of the study area, 64 withdrew consent, 16 were lost to follow up, and eight developed other physical conditions or problems. Subjects in the youngest age group had the largest number who did not complete all study activities (<u>Table 1</u>). Subclinical infections were only possible for per-protocol subjects since only this group had DENV HAI testing performed at 12 months. During 868 person-years of active surveillance in just the per-protocol subjects, 13 acute symptomatic DENV infections and 61 subclinical infections occurred.

The incidence of symptomatic DENV infection per 100 person-years in the overall cohort (including non-per-protocol subjects) was 1.62 (95%CI: 0.97, 2.58). Only one symptomatic infection occurred in subjects >15 years of age. The incidence of subclinical infection per 100

Age	Subjects, n ^a	Symptomatic DENV infection, n [n/100 person-yrs(95% Cl)] ^b	Subclinical DENV infection, n [n/100 person-yrs(95% Cl)] ^a	Total DENV infection, n/100 person-yrs	Ratio of subclinical to symptomatic DENV infection
6 mos —5 yrs	148	5 [2.50 (0.95, 5.49)]	15 [9.69 (5.66, 15.59)]	12.19	3.9:1
6–15 yrs	184	10 [4.90 (2.52, 8.70)]	19 [10.04 (6.25, 15.35)]	14.94	2.0:1
16–30 yrs	168	1 [0.5 (0.04, 2.31)]	12 [6.79 (3.71, 11.50)]	7.29	13.6:1
31–50 yrs	172	0 [0 (0, 1.32)]	7 [4.16 (1.85, 8.17)]	4.16	c
>50 yrs	182	0 [0 (0, 1.29)]	8 [4.46 (2.11, 8.42)]	4.46	d
All ages	854	16 [1.62 (0.97, 2.58)]	61 [7.03 (5.42, 8.96)]	8.65	4.3:1

Table 2. Incidence of symptomatic and subclinical dengue virus (DENV) infections in different age groups.

^aBased on 854 per-protocol subjects with both enrolment and 12-month blood collections.

^bBased on all subjects (per-protocol and non per-protocol)

^cSeven subclinical DENV infections and no symptomatic infections.

^dEight subclinical DENV infections and no symptomatic infections.

DENV = dengue virus; n = number; CI = confidence interval.

doi:10.1371/journal.pntd.0004337.t002

person-years in per-protocol subjects was 7.03 (95%CI: 5.42, 8.96), ranging from 4.16 in the 31–50 year group, to 10.04 in the 6–15 year group (<u>Table 2</u>). Three subjects were hospitalized for DHF, and five subjects received outpatient medical care (<u>S1 Table</u>), and no fatalities occurred.

DENV HAI seroprevalence increased sharply with age (Fig 2). The proportion of each age group that had a negative HAI profile at enrolment decreased from 56.1% in 6 month-5 year olds to 0.6% in 31.50 year olds, while the proportion having multitypic HAIs increased from 39.9% in 6 month-5 year olds to 98.8% in 31–50 year olds. The seroprevalence of multitypic HAIs was >98.3% for all ages >15 years.

The proportion of subjects who developed symptomatic DENV infection differed depending on baseline DENV HAI profile (Fig 3). Symptomatic infections occurred more often with monotypic (2 of 19) and negative (4 of 125) HAIs at enrolment, and less often with multitypic (7 of 710) HAIs (chi-square, p<0.001). In addition, the proportion of symptomatic infections among all DENV infections (i.e., symptomatic and subclinical) tended to differ depending on the HAI profile at enrolment (chi-square, p = 0.086) (Fig 4).

FOI was estimated by analyzing age-stratified DENV HAI seroprevalence at enrolment. Model 1, which assumed a constant FOI, yielded an average FOI of 0.044/year (95% CI: 0.039, 0.049) for each HAI serotype over the previous 20 years, with a log likelihood of -272 (nearly universal HAI seropositivity in adults older than 20 years did not allow FOI estimates beyond the previous 20 years). The model fit is shown in dark blue in Fig 5 with uncertainty shown in light blue. Model 2, which assumed non-constant FOI, yielded an average FOI of 0.061/year per serotype (95% CI: 0.051, 0.072) during the previous five years and 0.028/year per serotype (95% CI: 0.020, 0.036) for the 15 years before then, with a log likelihood of -265. The model fit is shown in red in Fig 5, with uncertainty shown in pink. Model 2 fit the data slightly better, particularly in the younger age groups, suggesting either that there was a higher FOI in the previous five years, or that subjects \leq 5 years of age experienced a higher FOI than older subjects. To assess whether different age groups may have had differing FOIs, DENV HAI seroconversions at 12 months in subjects who were dengue naive at enrolment were analyzed by age group (Fig 6). The age given in Fig 6 is the age at the start of the year in which seroconversions



Fig 2. Dengue virus (DENV) hemagglutination inhibition (HAI) profiles at enrolment in different age groups.

were considered. Because the number of dengue naive subjects was small (n = 125), there were large confidence intervals for these proportions. Nevertheless, there was no evidence of differential seroconversion in subjects \leq 5 years old than those >5 years, suggesting that the observed results from model 2 were due to FOI being higher during the previous five years than before then, rather than to differential exposure with age.

Discussion

In our study in a hyperendemic area of the Philippines, symptomatic DENV infections rarely occurred in subjects older than 15 years. Infections that did occur in this older age group were more likely to be subclinical than in those \leq 15 years. This age pattern of DENV infection occurred in the setting of high seroprevalence of multitypic DENV HAIs in adults likely due to high force of infection. This is one of the first prospective longitudinal cohort studies using active surveillance in which dengue incidence in adults and children have been determined within the same cohort.

The historical force of infection based on age-stratified seroprevalence at enrolment was found to be high in our study. Presuming a constant FOI, the infection rate among dengue naïve individuals was approximately 16% per year over the previous 20 years (based on FOI of 0.044/year per serotype). Presuming a non-constant FOI, the infection rate was about 22% per year during the previous five years (FOI of 0.061/year per serotype) and 11% per year before then (FOI of 0.028/year per serotype). Whichever of these two models was used, the results indicate that sustained DENV transmission has been ongoing in Cebu for many years. By the time subjects reached 16 years of age, over 96% had multitypic DENV HAIs. The high



Fig 3. Proportion of symptomatic dengue virus (DENV) infections among all per-protocol subjects with different DENV hemagglutination inhibition (HAI) profiles at enrolment. Negative HAI: 4/125; monotypic HAI: 2/19; multitypic HAI: 7/710 (chi-square, p<0.001). Three symptomatic cases in non-per-protocol subjects were not included because HAI testing was not completed.

population turnover in the Philippines leading to large numbers of dengue naïve individuals entering the population may be a contributing factor to the high FOI. In 2014, the birth and death rates in the Philippines were 25.34 and 5.02 per 1,000 population, respectively, as compared to 12.95 and 7.29 in Thailand (World Health Statistics 2014, World Health Organization). Nevertheless, other factors may also be playing a role in the Philippines, such as permissive environmental/ecological factors and/or incomplete vector control. The predominant viral serotype and strain during different years may have also affected the epidemic potential and, thus, FOI. The current epidemiology of dengue in the Philippines may mirror past epidemiology in other countries such as Thailand before demographic transition occurred.

We were unable to perform a sub-analysis of adult symptomatic dengue because so few symptomatic infections in adults occurred. This lack of cases existed in the setting of high seroprevalence of multitypic HAIs in adults. So while the independent effects of age and HAI profile on symptomatic dengue could not be evaluated, the lack of symptomatic infections in adults was likely related to the high seroprevalence of multitypic HAIs, which in turn was probably due to high historical FOI. It is notable that so few symptomatic infections occurred in subjects >15 years old despite a vigorous active surveillance system. The incidence of symptomatic dengue in children in our study was comparable to previous prospective cohort studies conducted in children in which the average rate has ranged from 0.6% to 3.9% per year [28], supporting the validity of our surveillance procedures for detecting dengue cases. Our findings demonstrate the primarily pediatric focus of dengue when FOI is high, and why dengue has been considered a childhood disease in many hyperendemic countries. Our study does not,



Fig 4. Proportion of symptomatic dengue virus (DENV) infections among total (symptomatic and subclinical) DENV infections in per-protocol subjects with different DENV hemagglutination inhibition (HAI) profiles at enrolment. Negative HAI: 4/12; monotypic HAI: 2/5; multitypic HAI: 7/57 (chi-square, p = 0.068). Three symptomatic cases were not included because of missing HAI titers.





doi:10.1371/journal.pntd.0004337.g005



Fig 6. Proportion of dengue naïve subjects at enrolment that seroconverted at 12 months.

however, clarify whether the clinical manifestations of dengue are more or less severe in adults since so few symptomatic adult cases were observed.

Symptomatic DENV infections occurred more frequently with baseline negative or monotypic HAIs than with multitypic HAIs. A multitypic HAI profile can result from two or more past DENV infections by different serotypes, although a single infection is still possible. If a monotypic HAI profile was presumed to have resulted from a single past infection, then our findings suggest that a second infection with a different serotype may be more likely to cause symptomatic infection.

There were several limitations to our study. First, only one year of surveillance data was available. Given the known cyclical nature of DENV transmission, other years may have different epidemiological and clinical patterns. For example, different predominating DENV serotypes or strains in the context of different population immunity in different years may lead to different ratios of symptomatic and subclinical disease [29]. Second, relatively few symptomatic infections occurred in the overall cohort limiting the statistical significance of our analyses. Third, the use of DENV HAI to determine both seroprevalence and seroconversion may have underestimated the actual rates given the unclear longevity of HAI titers. Measurement of more persistent markers of past infection would have been preferable, but were not performed due to resource limitations. It is, therefore, possible that the incidence of total DENV infections was actually higher than indicated by our study results. Nevertheless, the lack of symptomatic infections observed in adults would not have been affected.

In summary, our study demonstrates that symptomatic dengue is primarily a pediatric disease in hyperendemic areas with high force of infection. However, the average age of dengue could increase if force of infection decreases over time, which may be occurring in some dengue hyperendemic countries such as Thailand.

Supporting Information

S1 Table. Number of completed acute illness visits and follow up visits 1 after detection of acute febrile episodes (n = 274). (TIF)

S2 Table. Clinical presentation of symptomatic dengue virus (DENV) infections. (TIF)

S1 Checklist. STROBE checklist. (PDF)

S1 Dataset. Aggregated data. (XLSX)

Acknowledgments

The authors wish to acknowledge the contributions of Romelinda Goda Molabola and other clinical, laboratory, and administrative personnel at AFRIMS and PAVRU. We thank the medical staff at Punta Princesa health center and Cebu City Health Department for their support of the project. We thank Dr. Grace Tee Lewis for her epidemiological consultation. We are grateful to the children and adults involved in this study for their active participation. The views expressed in this article are those of the authors and do not represent the official policy or position of the U.S. Department of the Army, Department of Defense, or U.S. Government.

Author Contributions

Conceived and designed the experiments: MTA AS SF BT CK IKY. Performed the experiments: MTA AS CBL SF BT CK JMV AN IKY. Analyzed the data: MTA AS JMV CBL HEC SF JWL BT CK LRM AN LH IKY. Contributed reagents/materials/analysis tools: MTA AS JMV IATA CBL HEC SF JWL BT CK LRM AN LH DV IKY. Wrote the paper: MTA AS JMV IATA CBL HEC SF JWL BT CK LRM AN LH DV IKY.

References

- 1. Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, et al. (2013) The global distribution and burden of dengue. Nature 496: 504–507. doi: <u>10.1038/nature12060</u> PMID: <u>23563266</u>
- Gubler DJ (1997) Dengue and dengue hemorrhagic fever: Its history and resurgence as a global public health problem. In: Gubler DJ, Kuno G, editors. Dengue and dengue hemorrhagic fever. New York: CAB International. pp. 1–22.
- Special Programme for Research and Training in Tropical Diseases., World Health Organization. (2009) Dengue: guidelines for diagnosis, treatment, prevention, and control. Geneva: TDR: World Health Organization. 147 p. p.
- Grange L, Simon-Loriere E, Sakuntabhai A, Gresh L, Paul R, et al. (2014) Epidemiological risk factors associated with high global frequency of inapparent dengue virus infections. Front Immunol 5: 280. doi: <u>10.3389/fimmu.2014.00280</u> PMID: <u>24966859</u>
- Endy TP, Yoon IK, Mammen MP (2010) Prospective cohort studies of dengue viral transmission and severity of disease. Curr Top Microbiol Immunol 338: 1–13. doi: <u>10.1007/978-3-642-02215-9_1</u> PMID: <u>19802574</u>
- Gibbons RV, Kalanarooj S, Jarman RG, Nisalak A, Vaughn DW, et al. (2007) Analysis of Repeat Hospital Admissions for Dengue to Estimate the Frequency of Third or Fourth Dengue Infections Resulting in Admissions and Dengue Hemorrhagic Fever, and Serotype Sequences. Am J Trop Med Hyg 77: 910– 913. PMID: <u>17984352</u>

- Olkowski S, Forshey BM, Morrison AC, Rocha C, Vilcarromero S, et al. (2013) Reduced risk of disease during postsecondary dengue virus infections. J Infect Dis 208: 1026–1033. doi: <u>10.1093/infdis/jit273</u> PMID: <u>23776195</u>
- Yung CF, Lee KS, Thein TL, Tan LK, Gan VC, et al. (2015) Dengue Serotype-Specific Differences in Clinical Manifestation, Laboratory Parameters and Risk of Severe Disease in Adults, Singapore. Am J Trop Med Hyg.
- Leo YS, Thein TL, Fisher DA, Low JG, Oh HM, et al. (2011) Confirmed adult dengue deaths in Singapore: 5-year multi-center retrospective study. BMC Infect Dis 11: 123. doi: <u>10.1186/1471-2334-11-123</u> PMID: <u>21569427</u>
- Binh PT, Matheus S, Huong VT, Deparis X, Marechal V (2009) Early clinical and biological features of severe clinical manifestations of dengue in Vietnamese adults. J Clin Virol 45: 276–280. doi: <u>10.1016/j.</u> jcv.2009.04.004 PMID: <u>19451025</u>
- Kittigul L, Pitakarnjanakul P, Sujirarat D, Siripanichgon K (2007) The differences of clinical manifestations and laboratory findings in children and adults with dengue virus infection. J Clin Virol 39: 76–81. PMID: 17507286
- Mohd-Zaki AH, Brett J, Ismail E, L'Azou M (2014) Epidemiology of dengue disease in Malaysia (2000–2012): a systematic literature review. PLoS Negl Trop Dis 8: e3159. doi: <u>10.1371/journal.pntd.0003159</u> PMID: 25375211
- Limkittikul K, Brett J, L'Azou M (2014) Epidemiological trends of dengue disease in Thailand (2000– 2011): a systematic literature review. PLoS Negl Trop Dis 8: e3241. doi: <u>10.1371/journal.pntd.0003241</u> PMID: <u>25375766</u>
- Karyanti MR, Uiterwaal CS, Kusriastuti R, Hadinegoro SR, Rovers MM, et al. (2014) The changing incidence of dengue haemorrhagic fever in Indonesia: a 45-year registry-based analysis. BMC Infect Dis 14: 412. doi: 10.1186/1471-2334-14-412 PMID: 25064368
- Nisalak A, Endy TP, Nimmannitya S, Kalayanarooj S, Thisayakorn U, et al. (2003) Serotype-specific dengue virus circulation and dengue disease in Bangkok, Thailand from 1973 to 1999. Am J Trop Med Hyg 68: 191–202. PMID: 12641411
- Rodriguez-Barraquer I, Buathong R, Iamsirithaworn S, Nisalak A, Lessler J, et al. (2014) Revisiting Rayong: shifting seroprofiles of dengue in Thailand and their implications for transmission and control. Am J Epidemiol 179: 353–360. doi: <u>10.1093/aje/kwt256</u> PMID: <u>24197388</u>
- Cummings DA, lamsirithaworn S, Lessler JT, McDermott A, Prasanthong R, et al. (2009) The impact of the demographic transition on dengue in Thailand: insights from a statistical analysis and mathematical modeling. PLoS Med 6: e1000139. doi: 10.1371/journal.pmed.1000139 PMID: 19721696
- Morrison AC, Minnick SL, Rocha C, Forshey BM, Stoddard ST, et al. (2010) Epidemiology of dengue virus in Iquitos, Peru 1999 to 2005: interepidemic and epidemic patterns of transmission. PLoS Negl Trop Dis 4: e670. doi: 10.1371/journal.pntd.0000670 PMID: 20454609
- Porter KR, Beckett CG, Kosasih H, Tan RI, Alisjahbana B, et al. (2005) Epidemiology Of Dengue And Dengue Hemorrhagic Fever In A Cohort Of Adults Living In Bandung, West Java, Indonesia. Am J Trop Med Hyg 72: 60–66. PMID: <u>15728868</u>
- Siler JF, Hall MW, Hitchens AP (1926) Dengue: Its history, epidemilogy, mechanism of transmission, etiology, clinical manifestations, immunity, and prevention. The Philippine Journal of Science 29: 1– 304.
- Gubler DJ (1997) The emergence of dengue/dengue hemorrhagic fever as a global public health problem. In: Saluzzo JF, Dodet B, editors. Factors in the Emergence of Arbovirus Diseases. Paris: Elisevier. pp. 83–92.
- Bravo L, Roque VG, Brett J, Dizon R, L'Azou M (2014) Epidemiology of dengue disease in the Philippines (2000–2011): a systematic literature review. PLoS Negl Trop Dis 8: e3027. doi: <u>10.1371/journal.pntd.0003027</u> PMID: <u>25375119</u>
- Yoon IK, Alera MT, Lago CB, Tac-An IA, Villa D, et al. (2015) High rate of subclinical chikungunya virus infection and association of neutralizing antibody with protection in a prospective cohort in the Philippines. PLoS Negl Trop Dis 9: e0003764. doi: <u>10.1371/journal.pntd.0003764</u> PMID: <u>25951202</u>
- 24. Klungthong C, Gibbons RV, Thaisomboonsuk B, Nisalak A, Kalayanarooj S, et al. (2007) Dengue Viral Detection using Whole Blood for RT-PCR and Viral Isolation. J Clin Microbiol.
- Innis BL, Nisalak A, Nimmannitya S, Kusalerdchariya S, Chongswasdi V, et al. (1989) An enzymelinked immunosorbent assay to characterize dengue infections where dengue and Japanese encephalitis co-circulate. Am J Trop Med Hyg 40: 418–427. PMID: <u>2540664</u>
- Clarke D, Casals J (1955) Improved methods for hemagglutination studies with arthropod-borne viruses. Proc Soc Exp Biol Med 88: 96–99. PMID: 14357353

- Rodriguez-Barraquer I, Cordeiro MT, Braga C, de Souza WV, Marques ET, et al. (2011) From re-emergence to hyperendemicity: the natural history of the dengue epidemic in Brazil. PLoS Negl Trop Dis 5: e935. doi: <u>10.1371/journal.pntd.0000935</u> PMID: <u>21245922</u>
- Endy TP (2014) Human immune responses to dengue virus infection: lessons learned from prospective cohort studies. Front Immunol 5: 183. doi: <u>10.3389/fimmu.2014.00183</u> PMID: <u>24795725</u>
- 29. Endy TP, Anderson KB, Nisalak A, Yoon IK, Green S, et al. (2011) Determinants of inapparent and symptomatic dengue infection in a prospective study of primary school children in Kamphaeng Phet, Thailand. PLoS Negl Trop Dis 5: e975. doi: 10.1371/journal.pntd.0000975 PMID: 21390158